Author’s response to reviews

Title: The effect of gender and age on the association between weight status and health-related quality of life in Australian adolescents

Authors:

Kristy A Bolton (kabolton@deakin.edu.au)
Peter Kremer (peter.kremer@deakin.edu.au)
Naomi Rossthorn (naomi.rossthorn@outlook.com)
Marj Moodie (marj.moodie@deakin.edu.au)
Lisa Gibbs (lgibbs@unimelb.edu.au)
Elizabeth Waters (ewaters@unimelb.edu.au)
Boyd Swinburn (boyd.swinburn@auckland.ac.nz)
Andrea de Silva (Andrea.deSilva@dhsv.org.au)

Version: 2 Date: 10 July 2014

Author’s response to reviews:

Please see point by point response letter attached.